Author's response to reviews

Title: Breakthrough Reactions of Iodinated and Gadolinium Contrast Media after Oral Steroid Premedication Protocol

Authors:

Akiko Jingu (akiko_jingu@yahoo.co.jp)
Junya Fukuda (jfuku@gunma-u.ac.jp)
Ayako Taketomi-Takahashi (ayakorad@med.gunma-u.ac.jp)
Yoshito Tsushima (tyoshito@showa.gunma-u.ac.jp)

Version: 3
Date: 5 July 2014

Author's response to reviews: see over
July 6, 2014

Dear Dr. Danica Stephanie Lorrein Jose;

Re: Manuscript reference No. 1380561375103417

Please find attached a revised version of our manuscript “Breakthrough Reactions of Iodinated and Gadolinium Contrast Media after Oral Steroid Premedication Protocol” by Akiko Jingu et al., which we would like to resubmit for publication as a original article in BMC Medical Imaging.

Your comments and those of the reviewers were highly insightful and enabled us to greatly improve the quality of our manuscript.

In the following pages are our point-by-point responses to each of the comments of the reviewers as well as your own comments.

We also altered the format of the references. The reference numbers in the main text are now in square brackets. We also revised the order of the references, and deleted some irrelevant references. The captions for the tables are in the main manuscript file, following the references. We changed “Material and methods” in the abstract to read “Materials and methods.”

We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Akiko Jingu, MD
Dept. of Diagnostic Radiology and Nuclear Medicine Gunma University School of Medicine
3-39-22 Showa-machi, Maebashi, Gunma 371-8511 Japan
Tel) +81-27-220-8401
Fax) +81-27-220-8409
E mail) akiko_jingu@yahoo.co.jp